

Type IV pilus retraction enables sustained bacteremia and plays a key role in the outcome of meningococcal sepsis in a humanized mouse model

Jean-Philippe Barnier, Daniel Euphrasie, Olivier Join-Lambert, Mathilde Audry, Sophia Schonherr-Hellec, Taliah Schmitt, Sandrine Bourdoulous,

Mathieu Coureuil, Xavier Nassif, Mohamed El Behi

▶ To cite this version:

Jean-Philippe Barnier, Daniel Euphrasie, Olivier Join-Lambert, Mathilde Audry, Sophia Schonherr-Hellec, et al.. Type IV pilus retraction enables sustained bacteremia and plays a key role in the outcome of meningococcal sepsis in a humanized mouse model. PLoS Pathogens, 2021, 17 (2), pp.e1009299. 10.1371/journal.ppat.1009299. hal-03452015

HAL Id: hal-03452015 https://hal.science/hal-03452015v1

Submitted on 26 Nov 2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Citation: Barnier J-P, Euphrasie D, Join-Lambert O, Audry M, Schonherr-Hellec S, Schmitt T, et al. (2021) Type IV pilus retraction enables sustained bacteremia and plays a key role in the outcome of meningococcal sepsis in a humanized mouse model. PLoS Pathog 17(2): e1009299. https://doi. org/10.1371/journal.ppat.1009299

Editor: Christoph Tang, University of Oxford, UNITED KINGDOM

Received: September 28, 2020

Accepted: January 11, 2021

Published: February 16, 2021

Copyright: © 2021 Barnier et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: This work was supported by Université de Paris, Institut National de la Santé et de la Recherche Médicale (INSERM), Centre National de la Recherche Scientifique (CNRS), Fondation pour la Recherche Médicale (FRM) and a grant from Agence nationale de la Recherche (ANR) RETRACTOPATH (ANR-18-CE15-0019-01 - grant RESEARCH ARTICLE

Type IV pilus retraction enables sustained bacteremia and plays a key role in the outcome of meningococcal sepsis in a humanized mouse model

Jean-Philippe Barnier ^{1,2,3¤a}*, Daniel Euphrasie^{1,2}, Olivier Join-Lambert ^{1,2,3¤b}, Mathilde Audry^{1,2}, Sophia Schonherr-Hellec^{1,2}, Taliah Schmitt⁴, Sandrine Bourdoulous ^{1,5}, Mathieu Coureuil ^{1,2}, Xavier Nassif^{1,2,3®}*, Mohamed El Behi ^{1,2®}

1 Université de Paris, Faculté de Médecine, Paris, France, 2 Institut Necker Enfants-Malades, Inserm U1151, CNRS UMR 8253, Paris, France, 3 Service de microbiologie, Assistance Publique–Hôpitaux de Paris. Centre–Université de Paris, Hôpital Necker Enfants Malades, Paris, France, 4 Service de chirurgie reconstructrice et plastique, Groupe Hospitalier Paris Saint-Joseph, Paris, France, 5 Institut Cochin, Inserm U1016, CNRS UMR 8104, Paris, France

• These authors contributed equally to this work.

¤a Current address: Service de microbiologie, Assistance Publique–Hôpitaux de Paris. Centre–Université de Paris, Hôpital européen Georges Pompidou and Université de Paris, Centre de Recherche des Cordeliers, Inserm UMR-S 1138, Paris, France

¤b Current address: Service de microbiologie, CHU de Caen Normandie and Groupe de Recherche sur l'Adaptation Microbienne (GRAM 2.0, EA 2656), Université Caen Normandie, Caen, France * jean-philippe.barnier@inserm.fr (J-PB); xavier.nassif@inserm.fr (XN)

Abstract

Neisseria meningitidis (the meningococcus) remains a major cause of bacterial meningitis and fatal sepsis. This commensal bacterium of the human nasopharynx can cause invasive diseases when it leaves its niche and reaches the bloodstream. Blood-borne meningococci have the ability to adhere to human endothelial cells and rapidly colonize microvessels. This crucial step enables dissemination into tissues and promotes deregulated inflammation and coagulation, leading to extensive necrotic purpura in the most severe cases. Adhesion to blood vessels relies on type IV pili (TFP). These long filamentous structures are highly dynamic as they can rapidly elongate and retract by the antagonistic action of two ATPases, PiIF and PiIT. However, the consequences of TFP dynamics on the pathophysiology and the outcome of meningococcal sepsis in vivo have been poorly studied. Here, we show that human graft microvessels are replicative niches for meningococci, that seed the bloodstream and promote sustained bacteremia and lethality in a humanized mouse model. Intriguingly, although pilus-retraction deficient N. meningitidis strain ($\Delta pilT$) efficiently colonizes human graft tissue, this mutant did not promote sustained bacteremia nor induce mouse lethality. This effect was not due to a decreased inflammatory response, nor defects in bacterial clearance by the innate immune system. Rather, TFP-retraction was necessary to promote the release of TFP-dependent contacts between bacteria and, in turn, the detachment from colonized microvessels. The resulting sustained bacteremia was directly correlated with lethality. Altogether, these results demonstrate that pilus retraction plays a

recipient XN and SB). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

key role in the occurrence and outcome of meningococcal sepsis by supporting sustained bacteremia. These findings open new perspectives on the role of circulating bacteria in the pathological alterations leading to lethal sepsis.

Author summary

Invasive meningococcal diseases remain a major cause of fatal sepsis. A specific feature of Neisseria meningitidis is its ability to colonize the blood microvessels in a type IV pilus (TFP)-dependent mechanism. TFP are filamentous appendages that undergo retraction through a mechanism dependent on the PilT ATPase. Here, we assess the role of TFP retraction in the pathophysiology of meningococcal infection using a humanized model of SCID mice grafted with human skin. We show that human skin graft microvessels are replicative niches for N. meningitidis that promote sustained bacteremia and subsequent lethality. Intriguingly, although pilus retraction-deficient N. meningitidis (ΔpilT) efficiently colonizes human grafts, this mutant strain did not promote sustained bacteremia nor induce mouse lethality. This drastic decrease in virulence was not due to a decreased inflammatory response or to a defect in bacterial clearance by the innate immune system. Rather, we demonstrate that pilus retraction was crucial to obtain a sustained bacteremia by allowing the release of bacteria from colonized microvessels and that lethality was directly linked to sustained bacteremia. Altogether, these data demonstrate the important role of pilus retraction in meningococcal pathogenesis and infection outcome and open new questions regarding the impact of circulating bacteria on the host during lethal sepsis.

Introduction

Neisseria meningitidis (the meningococcus) is a Gram-negative extracellular bacterium whose ecological niche is the human nasopharynx [1]. In most cases, *N. meningitidis* is asymptomatically carried and lives in the nasopharyngeal mucus [2,3]. However, *N. meningitidis* is also responsible for invasive meningococcal diseases (IMD). The pathology is initiated when the bacterium crosses the nasopharyngeal epithelium, reaches the bloodstream and causes bacteremia leading to sepsis and meningitis. A central step in the pathophysiology of meningococcal infections is the extensive adhesion of meningococci to the endothelium in a process referred to as vascular colonization [4–7]. This colonization induces thrombotic events, deregulated inflammation and a leakage syndrome [6,8,9] leading to the development of *purpura fulminans* in the most severe forms [10]. At the early stage of IMD, clinical evidences shows that 42% to 70% of patients have cutaneous purpuric lesions that are known to contain viable meningococci [11–13]. At later time-points of infection, the massive disseminated microvascular colonization throughout the body in patients suffering from *purpura fulminans* is associated with a high bacterial load in the blood [14–16]. Moreover, vascular colonization is still present in the rare forms of chronic meningococcemia [17,18].

The capacity of pathogenic encapsulated meningococci to interact tightly with human endothelial cells relies on the expression of type IV pili (TFP) [19–21]. These long filamentous appendages are associated with many features such as adhesion to host cells, DNA uptake, twitching motility and bacteria-bacteria interactions [6,20,22–26]. Using infected SCID mice grafted with human skin, which recapitulates the tissue lesions observed in patients with IMD,

earlier works demonstrated that TFP are also required for meningococcal interaction with microvessels in vivo and subsequent vascular colonization, leading to tissue damage and lethality [6,20,27,28]. Neisserial TFP are composed of a core pilin subunit (PilE) and minor pilins (PilV, PilX, ComP) assembled into an helical structure [29,30]. Meningococcal TFP mediate bacterial aggregation, a process stabilized by the minor pilin PilX [24,31]. The biosynthesis of functional TFP involves a complex machinery comprising many proteins encompassing pilins and several accessory proteins [32]. Among them, the outer-membrane associated PilC proteins are encoded in N. meningitidis by pilC1 and pilC2 genes. PilC1 and PilC2 have similar but non-redundant functions in piliation. PilC1 is required for TFP adhesiveness, as a pilC1 mutant is piliated but non-adhesive [33-35]. PilC1 is proposed to be the tip-located TFP adhesin of N. meningitidis [36]. TFP are highly dynamic structures as they can elongate and retract through the antagonist action of two ATPases (PilF and PilT, respectively) that are associated with the inner membrane. PilC proteins are involved in TFP dynamics by antagonizing PilTmediated pilus retraction [34]. Moreover, PilT can downregulate the expression of PilC1 and negatively control the adhesiveness of the TFP by a mechanism independent of piliation [37]. Absence of pilus retraction, by deletion of the *pilT* gene, causes a hyper-piliated and hyperaggregative phenotype with increased adhesion to host cells whereas twitching motility and natural competence are lost [26,38]. In addition, cooperative retraction of bundled TFP enables nanonewton force generation that affects bacteria-bacteria and bacteria-host interactions [39]. In vivo, TFP retraction appears to be required for the adaptation of N. meningitidis to the constrained geometry of the microvessels [40]. However, whether TFP retraction plays a role in the pathophysiology and the outcome of meningococcal sepsis remains unknown.

Here, we took advantage of the validated model of SCID mice grafted with human skin to assess the role of TFP retraction in the pathogenesis of *N. meningitidis*. We demonstrate that although a PilT-deficient meningococcal strain ($\Delta pilT$, pilus retraction deficient strain) adhered and colonized human graft similarly to the wild-type (WT) strain, the mutant failed to promote a sustained bacteremia and mouse lethality. This effect was not due to reduced inflammatory response nor altered clearance of this mutant by the immune system. TFP retraction was necessary to trigger the release of TFP-dependent bacteria-bacteria contacts that enables bacteria to detach from colonized microvessels, a process required for blood seeding and sustained bacteremia. Furthermore, we show that lethality directly results from the ability of the WT strain to promote a sustained bacteremia. Altogether, these results demonstrate that pilus retraction plays a key role in the pathogenesis of *N. meningitidis* by promoting sustained bacteremia leading to lethal sepsis.

Results

Human graft is a replicative niche for Neisseria meningitidis

Previous work in a model of SCID mice grafted with human skin suggested that human microvessels provide nutritional niches for colonizing bacteria [5]. To further investigate the role of human skin graft as a replicative niche for blood-borne meningococci during systemic infection, we infected human-skin grafted mice intravenously (IV) with increasing inocula of *N. meningitidis*, from 10 up to 5x10³ colony forming units (CFU), and determined the ability of the bacteria to colonize the graft and proliferate. As few as 10 bacteria were sufficient to heavily colonize the graft (**Fig 1A**). For all inocula tested, the number of CFU recovered in human grafts 18 h post-infection (PI) were 3 to 4 orders of magnitude higher than the number of bacteria injected IV (**Fig 1A**). These results clearly show that in this model, meningococcal colonization of human skin graft is highly efficient and support massive bacterial multiplication.



Fig 1. Human graft is a replicative niche for *Neisseria meningitidis* **allowing sustained bacteremia**. **(A)** Grafted mice were infected with a rising range of inocula: 1×10^1 , 5×10^1 , 5×10^3 , and 5×10^3 CFU of WT *N. meningitidis*. Bacteremia and graft bacterial load were assessed at the time of sacrifice 18 h PI by quantitative culture. Bacterial counts are expressed in CFU/ml for blood and in CFU/g for graft. Two independent experiments, n = 5 or 6 mice per group, grafted with skin obtained from two different donors. Bars represent mean \pm SEM. **(B)** Grafted mice were infected intravenously with 5×10^6 CFU of WT *N. meningitidis* and its isogenic piliated non-adhesive mutant $\Delta pilCI$. Bacteremia was measured at 1, 18, 48 and 72 h by culturing serial dilutions of blood samples on agar plates. Two independent experiments, n = 6 or 8 mice per group. Bars represent mean \pm SEM, **** p < 0.0001, one-way ANOVA followed by multiple comparison test. **(C)** Kaplan-Meier plot showing the survival of infected grafted mice shown in panel B. Mice survival was assessed each day during 72 h. Two independent experiments, ** p < 0.01, two-sided log-rank Mantel-Cox analysis. **(D)** A competition assay between WT *N. meningitidis* and its isogenic piliated non-adhesive mutant $\Delta pilCI$ was performed on both grafted (left panel) and non-grafted (right panel) mice. Mice were infected IV with an inoculum of 1×10^7 CFU total (a mixture of 5×10^6 CFU of WT and 5×10^6 CFU of $\Delta pilCI$ mutant). Competitive index, defined as the mutant/WT ratio within the output sample, divided by the corresponding ratio in the inoculum, was measured in the blood of mice at 1 h and 18 h PI by tail vein blood puncture. Two independent experiments, n = 8 mice per group. Bars represent mean \pm SEM, NS p > 0.05; ** p < 0.01; **** p < 0.001, one-way ANOVA followed by multiple comparison test.

TFP-dependent vascular colonization is required to maintain a sustained bacteremia

We next aimed at determining the origin of the bacteria present in the bloodstream of infected grafted mice. We monitored the level of bacteremia in animals infected IV with $5x10^6$ CFU of WT *N. meningitidis*, a lethal inoculum in grafted mice [5]. As expected, 7 out of 8 mice had a bacteremia of 10^6 CFU/ml or above 18 h PI and died between 24 and 72 h. Only one mouse, which had a low bacteremia, survived (**Fig 1B and 1C**). All animals infected with a non-adhesive isogenic $\Delta pilC1$ strain survived and beyond 18 h exhibited a rapidly decreasing level of circulating bacteria (**Figs 1B and 1C and S1**). These results suggested that bacteria multiplying within the human graft. To test this hypothesis, we performed a competition assay using the WT adhesive strain and the non-adhesive $\Delta pilC1$ mutant. We reasoned that there would be an enrichment in adhesive WT bacteria within the bloodstream at 18 h PI due to the ability of WT to adhere within the human graft and seed the bloodstream. We enumerated the bacterial load of the WT and the non-adhesive $\Delta pilC1$ mutant-strain in the blood and defined the

competitive index as the mutant/WT ratio within the output sample, divided by the corresponding ratio in the inoculum. One-hour PI, in grafted mice infected with a mixture containing 5×10^6 CFU of WT and 5×10^6 CFU of $\Delta pilC1$ mutant, non-adhesive $\Delta pilC1$ bacteria showed an increased competitive index and were recovered at a greater number than WT bacteria (Fig **1D**, left panel). On the other hand, at 18 h PI, the competitive index decreased by an order of magnitude, indicating that the $\Delta pilC1$ mutant was poorly recovered from the blood compared to WT (Fig 1D, left panel). As a control, we performed a similar competition assay in nongrafted animals where neither WT nor $\Delta pilC1$ mutants are able to adhere to mouse blood vessels. Under these conditions, both strains were similarly cleared from the blood and no difference in the competitive index was observed at 1 h or 18 h PI (Fig 1D, right panel). The apparent competitive advantage of the $\Delta pilC1$ mutant at 1 h PI is likely a consequence of the high tropism of WT bacteria for the human graft and the removal of WT bacteria from the bloodstream through rapid adhesion to the human graft microvessels, while $\Delta pilC1$ mutants remain in the bloodstream. Conversely, over the course of the infection, at 18 h PI, the adhesive WT meningococci multiply actively within the graft and are released into the bloodstream, thereby outcompeting the non-adhesive $\Delta pilCI$ meningococci. Together, these results demonstrate that TFP-dependent colonization of the human vessels was a prerequisite to sustained bacteremia and that WT bacteria, but not the non-adhesive $\Delta pilCI$ meningococci, likely seed the blood by bacteria detaching from microcolonies grown within the human graft.

Pilus retraction is required to maintain a sustained bacteremia and to induce lethality

We next aimed to assess the role of pilus retraction in meningococcal pathogenesis using a pilus-retraction deficient mutant, $\Delta pilT$, and a complemented Cp-*pilT* strain expressing the WT *pilT* allele (Fig 2). Similar to the WT strain, the Cp-*pilT* strain killed 8 out of 9 (89%) of the infected mice within 24 to 48 h. Conversely, the $\Delta pilT$ strain killed only 1 out of 8 (12,5%) of the infected animals when injected at a similar dose (Fig 2B). Accordingly, while bacteremia was sustained in animals infected with the Cp-*pilT* strain, as observed in animals infected with the WT strain, bacteremia decreased by several orders of magnitude in $\Delta pilT$ mutant-infected mice (Fig 2A). This effect was not due to a colonization defect as graft colonization at 18 h PI reached 1x10⁸ CFU/g in mice infected with $\Delta pilT$ strain, similarly to what was observed in grafts colonized by the WT and Cp-*pilT* strains (Fig 2C). Together, these results indicate that pilus retraction is required to maintain a sustained bacteremia in human skin-grafted animals and to induce lethality.

N. meningitidis WT and $\Delta pilT$ trigger similar inflammatory responses and phagocytosis

We then addressed the question whether low bacteremia and higher survival observed with the $\Delta pilT$ mutant could result from a reduced inflammatory response and/or an increased clearance of this mutant by phagocytic innate immune cells. For this, we measured the level of murine and human pro-inflammatory cytokines in the blood of grafted mice infected with $5x10^6$ CFU of WT and $\Delta pilT$ at 4 and 18 h PI. Both WT and $\Delta pilT$ strains induced a canonical systemic inflammatory response with similar levels of murine IL-6, KC, TNF α and IL-1 β cytokines, as well as a similar vascular inflammation in the infected skin grafts, as measured by human IL-6, CXCL1, TNF α and IL-1 β cytokine levels (Fig 3A and 3B).

Innate immune responses play a major role in the control of IMD as the timeframe of the infection is not compatible with a preponderant role of the adaptative immune response [10,13,41]. While SCID mice are devoid of adaptive immunity, they have a functional innate



Fig 2. Pilus-retraction is required to establish sustained bacteremia and lethality. (A) Grafted mice were infected IV with $5x10^6$ CFU of *N. meningitidis*, $\Delta pilT$ mutant and its complemented Cp-*pilT* strain. Bacteremia of mice was measured at 1, 18, 48 and 72 h PI by culturing serial dilutions of blood samples on agar plates. Two independent experiments, n = 8 or 9 mice per group. Bars represent mean \pm SEM, * p < 0.05, one-way ANOVA followed by multiple comparison test. (B) Kaplan-Meier plot showing the survival of infected grafted mice shown in panel A. Mice survival was assessed each day during 72 h. Two independent experiments, ** p < 0.01, two-sided log-rank Mantel-Cox analysis. (C) Grafted mice were infected IV with $5x10^6$ CFU of WT *N. meningitidis*, isogenic $\Delta pilT$ mutant, or complemented strain Cp-*pilT*. Graft bacterial load at 18 hours PI was measured by quantitative culture on agar plates. Two independent experiments for WT and $\Delta pilT$ strains and one experiment for complemented Cp-*pilT* strain, n = 3 or 6 mice per group. Bars represent mean \pm SEM, NS p > 0.05, one-way ANOVA followed by multiple comparison test.

immune system. In particular, neutrophils, monocytes and red pulp macrophages (RPM) are present in the spleen of naïve SCID mice, although these mice lack marginal zone (MZ) macrophages and CD169⁺ metallophilic macrophages (S2 Fig) [42-44]. To assess the role of bacterial clearance in our model, we analyzed bacterial phagocytosis in different cell populations in the spleen and in the blood of infected animals by using a polyclonal antibody raised against N. meningitidis. We found that staining was barely detectable without permeabilization, indicating that the polyclonal antibody was specifically recognizing intracellular meningococci (S3 Fig). Upon infection with $5x10^6$ CFU of either WT or $\Delta pilT$ mutant in grafted SCID mice, phagocytosis of the $\Delta pilT$ mutant in the spleen by neutrophils, monocytes and RPM at 4 h and 18 h PI was not significantly different compared to that of the WT strain (Fig 4A). A similar result was observed in the blood where phagocytosis of WT and $\Delta pilT$ mutant by neutrophils and monocytes at 4 h and 18 h PI was equivalent (Fig 4A). In non-grafted SCID mice, phagocytosis of the two bacterial strains by blood and spleen phagocytes followed the same tendencies, with an exception for RPM phagocytosis of the $\Delta pilT$ mutant, which was slightly reduced as compared to that of WT at 4 h PI. However, this defect was not present at 18 h (Fig 4B). Hence, no significant difference was observed in the capacity of the innate immune cells of grafted mice to phagocyte the WT and $\Delta pilT$ mutant strains.



Fig 3. *N. meningitidis* WT and $\Delta pilT$ induce equivalent inflammatory responses (A, B) Serum levels of proinflammatory mouse and human cytokines were measured in grafted mice infected IV with 5×10^6 CFU of *N. meningitidis* WT strain or isogenic $\Delta pilT$ mutant at 4 h (A) and 18 h (B) post-infection using multiplex assays. Two independent experiments, n = 4, 5 or 6 mice per group. Bars represent mean \pm SEM, NS p > 0.05, unpaired t-test.

https://doi.org/10.1371/journal.ppat.1009299.g003



Fig 4. *N. meningitidis* WT and $\Delta pilT$ are similarly phagocytized. (A, B) *N. meningitidis* phagocytosis at 4 h and 18 h PI in the blood or spleen of grafted (A) and non-grafted (B) SCID mice. Neutrophils were identified as CD11b⁺Ly6G^{hi}. Red pulp macrophages (RPM) were identified as CD11b^{neg} F4/80^{hi} and proinflammatory monocytes were identified as CD11b⁺ Ly6G⁻ Ly6C^{hi} F4/80⁺. Two independent experiments for grafted mice with *n* = 6 mice in each group (B) and one experiment for non-grafted mice with *n* = 3 mice in each group (A). Bars represent mean ± SEM, NS *p* > 0.05; * *p* < 0.05, one-way ANOVA followed by multiple comparison test.

To further address the functional role of the innate immune system in bacterial clearance, SCID mice were treated with cyclophosphamide to deplete innate immune cells. Cyclophosphamide induced the depletion of neutrophils, monocytes, and splenic inflammatory monocytes but did not affect RPM (**S4 Fig**). Upon infection with a low inoculum of $5x10^3$ CFU of the WT strain, bacteremia at 18 h PI reached $6.0x10^3$ CFU/ml in non-treated mice whereas it reached $2.1x10^5$ CFU/ml, a 35-fold increase, in grafted animals treated with cyclophosphamide (**Fig 5A**). In contrast, bacteremia was significantly lower in animals infected with $5x10^3$ CFU of the $\Delta pilT$ strain, reaching $1.4x10^3$ CFU/ml in non-treated mice and $1.1x10^4$ CFU/ml, an 8-fold increase, in mice treated with cyclophosphamide (**Fig 5A**). Cyclophosphamide treatment also similarly increased bacterial loads in grafts of mice infected by the WT and $\Delta pilT$



Fig 5. Monocytes and neutrophils actively control the bacteremia induced by WT or $\Delta pilT$ strains. (A) Grafted and (B) non-grafted SCID mice treated with cyclophosphamide (cyclo) and infected with $5x10^3$ CFU of WT *N*. *meningitidis* or $\Delta pilT$ mutant. Bacteremia was assessed at 18 h post-infection by blood puncture. Mice were sacrificed at 72 h post-infection and graft bacterial load was measured by quantitative culture. Two independent experiments, with n = 6 mice in each group. Bars represent mean \pm SEM. NS p > 0.05; * p < 0.05, one-way ANOVA followed by multiple comparison test.

https://doi.org/10.1371/journal.ppat.1009299.g005

strains, reaching 1×10^7 to 1×10^8 CFU/g at 72 h PI (Fig 5A). In non-grafted control mice, bacteremia reached lower levels as compared to grafted mice, reflecting the need of human graft vessels for multiplication. Furthermore, no difference was observed in bacteremia between WT and $\Delta pilT$ infected non-grafted mice in the presence or in the absence of cyclophosphamide (Fig 5B). These data indicate that monocytes and neutrophils actively controlled the bacteremia of both WT or $\Delta pilT$ mutant in a similar manner.

Together, these data demonstrate that WT and $\Delta pilT$ strains triggered similar inflammatory responses and were similarly phagocytosed by phagocytic cells. This suggests that immune responses could not account for the low level of bacteremia and lethality observed in the $\Delta pilT$ -infected grafted mice. The data also demonstrate that sepsis-induced death is not solely the consequence of a systemic inflammatory response.

Pilus retraction is required for bacterial release from infected microvascular endothelium

The $\Delta pilT$ mutant is unable to retract TFP, is hyper-piliated and exhibits a hyper-aggregative phenotype *in vitro* [38,45]. We hypothesized that this might prevent the release of TFP-dependent contacts between bacteria and, in turn, the detachment process that releases bacteria back into the bloodstream to sustain bacteremia. To test this hypothesis, we first grew WT and $\Delta pilT$ strains *in vitro* in liquid medium under static condition and monitored by live microcopy the rate of bacterial aggregates formation and dispersal over a period of 20 h. As compared to a $\Delta pilX$ strain, an aggregation-defective mutant used as a negative control [24], both WT and $\Delta pilT$ strains displayed an aggregative phenotype. However, while within 10 h in culture the bacterial aggregates formed by the WT strain started to disaggregate, the $\Delta pilT$ aggregates were still growing and did not show signs of dispersal (Fig 6A). Similar to the WT strain, the complemented Cp-*pilT* strain displayed an aggregation/disaggregation phenotype, confirming that pilus retraction was required for bacteria to disaggregate.



Fig 6. PilT-dependent pilus retraction is critical for bacterial release from microcolonies adherent to endothelial cells. (A) Aggregates of WT *N. meningitidis*, its isogenic $\Delta pilT$ mutant, Cp-*pilT* complemented strain and $\Delta pilX$ mutant producing GFP under an IPTG-inducible promoter and grown in liquid medium were imaged using the IncuCyte S3 platform at 37°C with 5% CO₂. Four image sets from distinct regions per well were taken every 15 min using a ×20 dry objective and each condition was run in triplicate. Bacterial aggregates formation and dispersion was measured over a period of 20 h under static condition. (B) Number of bacteria detaching from microcolonies adhering onto human dermal microvascular cells (HDMEC) under flow condition. Cells were infected for 30 min under static condition with WT *N. meningitidis*, $\Delta pilT$ mutant, Cp-*pilT* complemented strain, and $\Delta pilC1$ mutant as non-adhesive control. Cells were then washed to remove non-adherent bacteria and placed under shear stress of 0.15 dyn/cm². Bacterial count in the effluent was evaluated each hour by plating serial dilutions on agar plates. Three independent experiments. Bars represent mean ± SEM. * p < 0.05, one-way ANOVA followed by multiple comparison test.

https://doi.org/10.1371/journal.ppat.1009299.g006

To determine whether the lack of pilus retraction might prevent bacterial release from infected microvessels, we assessed in vitro the bacterial detachment from microcolonies grown at the surface of primary human dermal endothelial cells (HDMEC). During bacterial growth, numerous bacteria were released in the medium from microcolonies formed by WT meningococci whereas very few bacteria detached from $\Delta pilT$ microcolonies. We quantified the number of bacteria detaching from microcolonies grown at the surface of HDMEC under shear stress conditions that mimic the blood flow observed in capillaries (0.15 dyn/cm²) [46]. After 5 h under shear stress, WT bacteria started to significantly release from the endothelial cell surface to reach 6.5×10^6 bacteria/ml at 7 h. Conversely, the $\Delta pilT$ mutant detached poorly from the infected monolayer under flow, reaching only 4.3x10⁵ bacteria/ml at 7 h, and this release was markedly delayed as compared to the WT strain (Fig 6B), despite having similar adhesion ability (S5 Fig). The complemented Cp-*pilT* strain behaved like the WT strain and as expected, no significant bacterial release was observed for the non-adhesive $\Delta pilC1$ control mutant, which is unable to colonize the endothelial monolayer (Fig 6B). These data demonstrate that $\Delta pilT$ bacteria poorly detach from adherent microcolonies formed on human endothelial cells and suggest that this defect also likely affects the dispersion mechanism required for a sustained bacteremia in vivo.

Maintenance of a sustained bacteremia correlates with lethality in infected non-grafted mice

Our data suggested that lethality was correlated with the ability of meningococcal strains to promote sustained bacteremia. To further investigate this correlation, we artificially maintained bacteremia above 1×10^6 CFU/ml for both WT or $\Delta pilT$ mutant over a period of 18 h in non-grafted mice by injecting 3×10^6 CFU twice with an interval of 8 h between injections (Fig 7A and 7B). Induced sustained bacteremia was associated with a high mortality rate between 18 h and 48 h in mice infected with both strains (lethality of 6/6 and 5/6 mice for the WT strain and the $\Delta pilT$ mutant, respectively) (Fig 7A and 7B). Hence, we establish a correlation between an artificially sustained bacteremia and lethality in mice regardless of bacterial strain used.

Taken together, our data demonstrate that pilus retraction is required to release TFPdependent bacterial contacts and promote a continuous sustained bacteremia responsible for mice lethality.

Discussion

TFP are major virulence factors that mediate adhesion of meningococci to human vessels and are involved in the triggering of tissue lesions. Here, we provide direct evidence that human graft microvessels confer a highly efficient replicative niche for adherent meningococci, that supports extensive bacterial multiplication. Following bacterial growth, pilus retraction is critical to release TFP-dependent bacterial contacts for bacteria to disseminate from this reservoir into the bloodstream to sustain bacteremia. As lethality was clearly associated with sustained bacteremia, we conclude that TFP retraction plays a key role in the pathophysiology and outcome of meningococcal sepsis.

TFP are widespread virulence factors among both Gram-negative and Gram-positive bacterial species, mediating physical interaction of bacteria with host cells and triggering signaling events that are essential in bacterial pathogenesis [32,47–49]. TFP biology is a complex system involving a large number of genes, many of which are conserved in both sequence and genomic organization in piliated species. In particular, PilT is highly conserved among TFPexpressing species and mediates the dynamic retraction of TFP. In many species, TFP



Fig 7. Maintenance of a sustained high bacteremia correlates with lethality. (A, B) Non-grafted SCID mice were infected IV with *N. meningitidis* WT strain (A) or $\Delta pilT$ mutant (B). A first infection was performed with an inoculum of $3x10^6$ CFU. A second infection with the same inoculum was performed in the 2 injections group 8 h after the first injection to mimic a prolonged bacteremia (2 injections), as control mice received one injection (1 injection). Mice survival was assessed each day during 72 h. Two independent experiments with *n* = 6 mice in each group. Kaplan-Meier plots showing the survival of infected mice (left panel, **p < 0.001; *** p < 0.001, two-sided log-rank Mantel-Cox analysis). Bacteremia was measured at 1, 8, 18, 48 and 72 h by culturing serial dilutions of blood samples on agar plates (right panel, bars represent mean ± SEM, **** p < 0.0001, one-way ANOVA followed by multiple comparison test).

retraction has been shown to control motility behavior, important properties of bacterial aggregates and microcolony formation [48]. Different studies have shown a role for PilT in virulence for several human or veterinary bacterial pathogens, including Escherichia coli, Pseudomonas aeruginosa, Legionella pneumophila or Dichelobacter nodosus [50-54]. In enteropathogenic E. coli, disruption of the ATP binding site of BfpF, a presumed functional homolog of PilT, reduces dispersion of bacterial aggregates and leads to a marked decrease in infectivity [50]. In L. pneumophila, an analogous change in DotB, a PilT family member based on sequence similarity, prevents the survival within macrophages [54]. Consistent with a role for pilus retraction in motility and intimate cell adhesion, a P. aeruginosa PilT-deficient strain is significantly less infectious than the WT strain in mouse models of corneal infection and acute pneumonia [51]. The *P. aeruginosa* $\Delta pilT$ mutant colonized the lungs as well as the parental strain, but exhibited a reduced ability to disseminate from the lung to the liver [51]. Few data are available for the impact of pilus-retraction on N. meningitidis virulence in vivo. Transgenic mice expressing human CD46, a proposed adhesion receptor for *N. meningitidis*, and infected with a $\Delta pilT$ mutant displayed a reduced bacteremia and a higher survival rate compared to the WT strain [52]. Nevertheless, the adhesion and virulence of meningococcus in this model is not TFP-dependent [55,56].

In *Neisseria* species, earlier studies performed *in vitro* have shown that inactivation of *pilT* gene results in a hyper-piliated state that is a consequence of the lack of TFP retraction. This phenotype increases binding to host cells *in vitro* [34,37,38,45]. Pilus-retraction promotes

intimate adhesion leading to spreading of microcolonies onto a cell monolayer and is responsible for the full activation of the β 2-adrenergic receptor that functions as a signaling receptor in infected endothelial cells [38,57,58]. In addition, it was shown that TFP retraction promotes intermittent TFP-dependent traction forces inside bacterial aggregates that control important properties of microcolonies, including viscosity, local order and shape [59,60]. In vitro, this property allows N. meningitidis aggregates to fill microfluidic channels, to adjust to constrained geometry and to spread onto a cell monolayer [38,40]. It has been proposed that this feature may allow meningococci to efficiently colonize the human microvasculature and to provoke capillary occlusions [40]. We show here that while the shape of the colonies formed within microvessels may vary between WT and retraction deficient stains, the number of bacteria that colonized the human vessels in vivo were similar, indicating that TFP retraction has no major impact on meningococcal adhesion and proliferation in vivo. In addition, colonization was associated with the production of equivalent cytokine levels indicating that both strains induced similar systemic and vascular inflammation. Our main observation was that loss of pilus retraction affects bacterial detachment from adherent microcolonies grown on endothelial cells, thereby reducing bacterial dissemination. These results provide evidence that TFP retraction is an essential process to release TFP-dependent contacts within bacterial microcolonies formed at the endothelial cell surface and is required for bacterial dissemination and seeding of the bloodstream. Whether the hyper-piliated state of the $\Delta pilT$ mutant may also participate per se to the lower release of bacteria observed in our in vivo model remains an open question. To address this point would require a strain that is as piliated as the $\Delta pilT$ mutant, but still able to retract pili. To our knowledge, there is no artificial means to upregulate the piliation to the level of a $\Delta pilT$ strain without affecting retraction.

Besides TFP, several other adhesins that are playing a role in meningococcal adhesion to host cells have been described. Among them, Opa and Opc are abundant in the outer-membrane and the best studied [61]. Unlike TFP, they are not expressed in all invasive isolates and are partially masked by the capsule, adhesion of encapsulated meningococci therefore relies on TFP [62–64]. In this study, we used Opa⁻ and Opc⁻ isolates in order to avoid interferences with these adhesins. However, *in vitro* and clinical data suggest a role for Opc in the pathogenesis of meningococcal disease. Opc has been shown to promote cell invasion and transcytosis across the brain endothelial barrier [65–67]. Interestingly, *N. meningitidis* strains of the hyperinvasive ET-37/ST11 clonal complex that lack the *opc* gene have been reported to cause severe cases of septicemic IMD but have a relatively low tendency to cause meningitis [68,69]. Overall, while Opc may enhance the ability to cause meningitis, this bacterial factor is not likely to play a preponderant role in vascular colonization and during the bloodstream phase of the disease.

So far, the mechanisms leading to bacterial detachment and dissemination have been poorly studied. Factors involved in bacterial adhesiveness are good candidates to influence bacterial detachment and dissemination. Among them, PilC proteins are subjected to phase variation, a mechanism enabling rapid adaptation to environmental selective pressure [70]. Because *pilC* mutants express low levels of TFP and are less adhesive, phase variation-mediated switching to OFF of PilC expression may be a means by which meningococci could detach from endothelial cells and reach the bloodstream. Genomic studies have shown that the absence of *pilC* genes expression was more frequently found in invasive rather than carriage isolates, for example in *N. meningitidis* serogroup W ST-11 clonal complex strains [71,72]. *In vitro*, it was shown that following adhesion to epithelial cells, a posttranscriptional modification of the major pilin PilE can occur that destabilizes TFP-dependent contacts and promotes detachment of meningo-cocci from adherent microcolonies [73]. More recently, it was shown that lactate released by epithelial cells upon meningococcal adhesion, can act as a signal molecule that induces dispersion of bacterial microcolonies [74]. In the gonococcus, oxygen depletion promotes TFP

retraction-driven microcolony disassembly *in vitro* [75]. Whether these mechanisms play a role in bacterial dissemination and disease *in vivo* remains to be determined. Here, we demonstrate that bacterial release that is promoted by TFP retraction plays a major role in maintaining sustained bacteremia. The impaired ability to maintain a high bacteremia in animals infected by the $\Delta pilT$ mutant was responsible for a striking decrease in virulence, as all animals survived infection. When bacteremia of the $\Delta pilT$ mutant was artificially maintained in non-grafted mice, it induced a similar lethality pattern as the one induced by the WT strain, indicating that a sustained high bacteremia can be solely responsible for lethal sepsis. This observation is consistent with clinical reports showing that the severity of IMD is linked to the amount of circulating bacteria in the blood [10,76].

A major question arising from this work is why and how sustained bacteremia is required to promote lethality. We subsequently addressed the role of the immune response in the control of bacteremia and lethality. The acute nature of IMD is not temporally compatible with the development of an effective adaptative immune response [10,13]. Hence, the innate immunity plays a major role in the control of meningococcal infections. This is highlighted by clinical evidences showing that immunodeficiencies increasing the risk of IMD in humans are mostly innate immune deficiencies (i.e. defects in complement, defects in inflammatory response, or spleen dysfunction) [77,78]. The human skin-grafted SCID mice therefore represent a pertinent model, since these mice have all the key effectors of the innate immune response (inflammation, complement, phagocytes, NK cells). Interestingly, we observed that infection by both WT and $\Delta pilT$ strains led to similar systemic and vascular inflammatory responses. This is consistent with their ability to similarly colonize the vascular wall and likely produce equivalent levels of pathogen-associated molecular patterns (PAMPs) such as the lipo-oligosaccharide (LOS), the pro-inflammatory endotoxin [79]. Hence, in our experimental conditions, a deregulated host inflammatory response was not correlated with the outcome. Until recently, the magnitude of the LOS-induced cytokine response was commonly used to assess immune function in patients undergoing critical sepsis. Plasma levels of pro-inflammatory cytokines such as TNF α , IL-1 β , IL-6 and IL-8 were correlated with disease outcome [10,80–82]. However, the exact role of these cytokines in the pathogenesis of sepsis was difficult to define since their activity is modulated by the simultaneous activation of compensatory anti-inflammatory pathways shortly after sepsis initiation [83]. The hallmark cytokine is IL-10, which is produced by a variety of leukocytes, suppressing the production of IL-6 and interferon- γ (IFN γ) and stimulating the production of soluble TNF receptor and IL-1 receptor antagonists. Finally, attempts to improve survival by dampening the inflammatory response by way of IL-1 and TNF blockade had little success in reducing mortality [83], further supporting the notion that increased systemic inflammation cannot solely account for organ failure and mortality. In addition, we did not find any major differences in the cellular innate immune response to infection by the WT and $\Delta pilT$ strains likely to explain the stark contrast between these two strains in promoting a lethal outcome. These results indicate that, beside an overwhelming inflammation, sustained bacteremia is required to promote organ dysfunction and death. One limitation of our model is the restriction of bacterial adhesion to the human graft vasculature as the bacteria cannot interact with the microvasculature of some important organs (kidney, heart, adrenal glands) that are targets of blood-borne meningococci in patients with IMD and participate in multiple organ failures and lethality [84–86]. It is, however, likely that bacterial persistence in the bloodstream provokes profound circulatory, cellular, and/or metabolic abnormalities, associated with a greater mortality. Indeed alteration in redox balance, calcium homeostasis, endothelial integrity, energy metabolism in monocytes can all play crucial roles in sepsis progression [87-89]. Future work will aim at a better definition of the impact of sustained bacteremia in the cellular and molecular dysfunctions leading to lethal sepsis.

Altogether, this work reveals the key role of TFP retraction in the occurrence and outcome of lethal meningococcal sepsis by facilitating the release of meningococci from the vascular niche. This work furthermore highlights the role of sustained bacteremia in the pathological alterations leading to lethal sepsis.

Material and methods

Ethics statement

The animal experimental procedures described in this paper are in accordance with the European ethical legislation (Directive 2010/63/EU). The experimental protocol was approved by the Comité d'Expérimentation Animale de l'Université Paris Descartes (project number 12–030 and 2018012515596498). Human skin grafts were obtained from surgical waste from patients undergoing plastic surgery at Groupe Hospitalier Paris Saint-Joseph (Paris, France). According to the French legislation, the patients were informed of the research purpose and their non-opposition was orally received.

Bacterial strains and growth conditions

N. meningitidis 8013 strain clone 2C4.3 is a piliated, adherent and encapsulated serogroup C clinical isolate that produces the class I SB pilin variant, Opa⁻, Opc⁻, PilC1⁺/PilC2⁺, as described previously [19]. The $\Delta pilT$ mutant was engineered by introduction of an *aph3'* kanamycin resistance cassette into the pilT gene locus. Chromosomal sequences upstream and downstream flanking regions of the *pilT* gene were amplified using PilT_Fr1_Fw 5'-TCAGGA TGAAGTCTTGGATGG-3' and PilT_Fr1_Rv 5'-TCAGCTCATTCACACAACCGCCTTCCG GCCATACC-3' for upstream region and PilT_Fr2_Fw 5'-GTTCTTCTGAAATGCGGCTCT GTTTAGTATAATG-3' and PilT_Fr2_Rv 5'-CGTCTCAATCAAAGGTTTGCCGTC-3' primers for downstream region. PilT_Fr1_Rv and PilT_Fr2_Fw primers contained 10 nucleotides overlapping regions with the *aph3'* kanamycin-resistance cassette at their 3' and 5' end, respectively. The kanamycin-resistance cassette was amplified with PilT Kn Fw 5'-CGGTTGTGT GAATGAGCTGATTTAACAAAAATTTAAC-3' and PilT_Kn_Rv 5'-GAGCCGCATTTCA GAAGAACTCGTCAAGAAGGCGATAG-3' primers that contained 10 nucleotides overlapping regions with upstream and downstream flanking regions at 5' and 3' end, respectively. This construct was then assembled by overlapping PCR and transformed into 2C4.3 WT strain. A *pilT* complemented strain (designated Cp-*pilT*) expressing a WT *pilT* allele under the control of the strong constitutive opaB promotor from N. gonorrhoeae was engineered by amplifying the *pilT* allele with PilT_PacI_Fw 5'-GGCCTTAATTAAGGAGTAATTTTATG CAGATTACCGACTTACTCGCCTTCGGC-3' and PilT_XbaI_Rv 5'-CCGGTCTAGATCA GAAACTCATACTTTCGC-3' primers that contained overhang restriction sites for PacI and XbaI. This PCR fragment was restricted and cloned between the PacI and XbaI sites of the pMR32 plasmid encoding an erythromycin resistance marker ermC [90]. The complementation construct was then inserted into 2C4.3 WT strain between trpB and iga genes by transformation. The $\Delta pilT$ mutant kanamycin resistant construct was then introduced into the chromosome of the *pilT* complemented strain by homologous recombination. N. meningitidis $\Delta pilX$ and $\Delta pilC1$ mutants were described elsewhere [24,33]. Bacterial strains expressing the green fluorescent protein (GFP) under the control of an IPTG-inducible promoter were obtained by transformation with the pAM239 plasmid [46]. Bacterial strains were stored frozen at -80°C and routinely grown on gonococcal base (GCB) agar plates (Difco) containing Kellogg's supplements at 37°C in moist atmosphere containing 5% CO₂. N. meningitidis strains were selected using kanamycin at 100 mg/l and erythromycin at 3 mg/l. To select E. coli DH5α strains used for plasmid propagation, kanamycin was used at 20 mg/l.

Mouse model of infection

Six to 8-week-old CB17/Icr-Prkdcscid (Severe Combined Immunodeficiency: SCID) female mice were obtained from Janvier Labs (Saint-Berthevin, France). Mice were grafted with normal human skin as previously described [6]. Briefly, human skin was obtained from surgical wastes from patients undergoing plastic surgery. After removing adipose tissue, full thickness human skin (dermis and epidermis) was grafted onto the back of SCID mice by surgical stitching after an intraperitoneal anesthesia with ketamine 100 mg/kg, xylazine 10 mg/kg and buprenorphine 0,1 mg/kg. After a few weeks, human graft microvessels spontaneously connect with murine circulation, enabling the presence of endothelial cells of human nature in mice. Mice were infected with N. meningitidis strains grown overnight at 37°C on GCB agar plates prepared without iron (Kellogg's supplement II) and supplemented with deferoxamine (Desferal, Novartis) at a final concentration of 15 μ M. Bacterial colonies were harvested and cultured in RPMI with 1% bovine serum albumin and $0.06 \,\mu$ M deferoxamine under gentle agitation to reach the exponential phase of growth. Bacteria were then resuspended in physiological saline. Mice were infected intravenously and 10 mg of human holotransferrin (R&D Systems) was administered intraperitoneally just before infection. Mice that have developed signs of lethal disease were euthanized. Bacteremia was assessed by tail vein puncture. Skin-graft bacterial load was measured by crushing and homogenizing the tissue with Lysing Matrix M tubes and Fast-Prep (MP Biomedicals). Bacterial counts were determined by plating serial dilutions of the samples onto GC agar plates. The bacteria were vortexed thoroughly before plating to break up aggregates. To deplete macrophages and neutrophils, mice were treated with cyclophosphamide (C7397, Sigma-Aldrich) resuspended in physiological saline administered intraperitoneally. A first dose of 150 mg/kg was administered 4 days prior infection, a second dose of 100 mg/kg the day before infection and a third dose of 100 mg/kg 2 days post-infection when necessary.

Bacterial detachment assay

HDMEC primary endothelial cells (Promocell) were grown in laminar flow chambers composed of six independent flow channels (μ -Slide VI 0.4, Ibidi) coated with rat tail type I collagen under 5% CO₂ until confluence. Bacteria were grown on GCB agar plates, adjusted to OD₆₀₀ = 0.02 in prewarmed endothelial cell medium (ECM) and cultivated for 2 hours at 37°C under shaking conditions. Cells were infected with 10⁶ bacteria (multiplicity of infection = 100) and bacteria were allowed to adhere to the cells for 30 min, then unbound bacteria were removed by three extensive washes. The number of bacteria adhering onto HDMEC at 30 min was counted after DAPI staining. Infected cells were then connected to a continuous flow of ECM applying a shear stress of 0.15 dynes/cm² using a syringe pump (Harvard Apparatus). The flow chamber was placed in an incubator at 37°C with 5% CO₂ throughout the experiment. Every hour, samples coming out of the flow chamber were collected, serial dilutions were performed and cultured on GCB agar plates to determine CFU.

Flow cytometry

Blood was collected from deeply anesthetized mice by cardiac puncture using heparinized syringes. Spleens were removed and passed through a 70-µm filter. Red blood cells were lysed using RBC lysing buffer (Sigma). Cells were blocked using anti-CD16/32 antibody (2.4G2 clone; Bio X Cell) for 20 min on ice, then stained for 20 min with fluorescently labeled antibodies including FITC-Ly6C (HK 1.4), PE-F4/80 (BM8), PerCP Cy5.5-Ly6G (1A8), APC-Cy7-CD11b (M1/70), Brilliant Violet 421-CD45 (30-F11), and Brilliant violet 510-MHC-II (M5/114). Nonviable cells were identified using BD Horizon Fixable viability Stain. After surface staining, cells were fixed and permeabilized using Fix & Perm Cell Permeabilization kit according to manufacturer instructions (Life technologies). Phagocytosis of WT and $\Delta pilT$ mutant were detected by intracellular staining using a rabbit polyclonal IgG against *N. menin-gitidis* [19] and conjugated to allophycocyanin (APC) according to manufacturer instructions (Life technologies). Isotype control antibodies were used to confirm positive signals. All antibodies were purchased from Biolegend. Neutrophils were identified as CD11b⁺Ly6G^{hi}. Red pulp macrophages were identified as CD11b^{neg} F4/80^{hi} and proinflammatory monocytes were identified as CD11b⁺ Ly6G⁻ Ly6C^{hi} F4/80⁺.

Aggregates formation and dispersion assay

GFP expressing bacteria were grown in FluoBrite DMEM (Thermofisher Scientific) containing 1 mM IPTG. Bacterial aggregates were imaged using the IncuCyte S3 system (Essen Biosciences) housed within an incubator at 37° C with 5% CO₂. Images from phase contrast and green (400-ms exposure) channels were acquired every 15 min for 18 hours using a ×20 dry objective. Four set of images were acquired per condition, and each condition was run in triplicate. Analysis and quantification were performed using IncuCyte S3 Software (Essen Biosciences).

Multiplex cytokines immunoassays

Mice blood was centrifuged at 4000 g for 20 min at 4 °C and sera were collected and frozen at -80 °C. The concentrations of human and murine cytokines were quantified by electrochemiluminescence using multiplex assay kits from Meso Scale Discovery. Briefly, 25 µl of serum were added to the 96-well multi-array plates and the assays were processed following the manufacturer's instructions. Plates were read on the multiplexing imager Sector S600 (Meso Scale Discovery).

Statistical analysis

Statistical test is specified for each figure, p values of < 0.05 were considered statistically significant. Statistical analysis was performed using GraphPad Prism 8 software.

Supporting information

S1 Fig. Colonization of human skin graft by *N. meningitidis* WT and $\Delta pilC1$. Grafted mice were infected IV with 5×10^6 CFU of WT *N. meningitidis* and isogenic $\Delta pilC1$ mutant. Graft bacterial load at 4 hours PI was measured by quantitative culture on agar plates Two independent experiments, n = 6 mice per group. Bars represent mean \pm SEM, ** p < 0.05, unpaired t-test. LOD: limit of detection.

(TIF)

S2 Fig. Gating strategy used to identify splenic immune cells. Representative flow cytometry plots identifying splenic neutrophils (CD11b⁺Ly6G^{hi}), red pulp macrophages, RPM (CD11b^{neg} F4/80^{hi}) and inflammatory monocytes (CD11b⁺ Ly6G⁻ Ly6C^{hi} F4/80⁺) in the spleen of SCID mice.

(TIF)

S3 Fig. Polyclonal rabbit antibody raised against *N. meningitidis* recognize efficiently phagocytized bacteria. Blood neutrophils from mice infected with WT *N. meningitidis* or $\Delta pilT$ mutant were permeabilized or not (as described in the material and methods section) and stained with an APC-conjugated rabbit polyclonal IgG against *N. meningitidis*. (A)

Representative flow cytometry plots showing efficient staining in permeabilized cells. (B) Quantification of the staining efficiency in permeabilized and non-permeabilized neutrophils. Two independent experiments with n = 4 mice per group, * p < 0.05, two-tailed Mann-Whitney test.

(TIF)

S4 Fig. Cyclophosphamide treatment depletes neutrophils, monocytes, and splenic inflammatory monocytes in SCID mice but do not affect RPM. Mice were treated with PBS or cyclophosphamide (cyclo) as described in the method section. After 24 h, blood and spleens were collected and analyzed by flow cytometry for the presence of neutrophils, monocytes and RPM. (A) Representative flow cytometry plots showing efficient depletion of neutrophils, monocytes, and splenic inflammatory monocytes after cyclophosphamide treatment. (B) Quantification of cell depletion using absolute cell counts. One experiment, n = 2 mice in each group.

(TIF)

S5 Fig. Adhesion of *N. meningitidis* WT and $\Delta pilT$ on endothelial cells. Number of bacteria adhering onto human dermal microvascular cells (HDMEC) at 30 min under shear stress of 0.15 dyn/cm². Cells were infected for 30 min with WT *N. meningitidis* and $\Delta pilT$ mutant and adhesive bacteria were counted after DAPI staining. Three independent experiments. Bars represent mean ± SEM, NS *p* > 0.05, unpaired t-test. (TIF)

Acknowledgments

We express our gratitude to Pamela Schnupf for careful reading of the manuscript. We thank Karine Bailly of the Institut Cochin Cytometry and Immunobiology facility (CYBIO) for excellent technical assistance and the Institut Necker Enfants Malades histology facility for technical support.

Author Contributions

Conceptualization: Jean-Philippe Barnier, Olivier Join-Lambert, Xavier Nassif, Mohamed El Behi.

Formal analysis: Jean-Philippe Barnier, Mohamed El Behi.

Funding acquisition: Sandrine Bourdoulous, Xavier Nassif.

Investigation: Jean-Philippe Barnier, Daniel Euphrasie, Mathilde Audry, Sophia Schonherr-Hellec, Mohamed El Behi.

Methodology: Jean-Philippe Barnier, Mathieu Coureuil, Mohamed El Behi.

Project administration: Xavier Nassif.

Resources: Taliah Schmitt.

Supervision: Olivier Join-Lambert, Xavier Nassif.

Validation: Jean-Philippe Barnier, Mohamed El Behi.

Writing - original draft: Jean-Philippe Barnier, Xavier Nassif, Mohamed El Behi.

Writing – review & editing: Sophia Schonherr-Hellec, Sandrine Bourdoulous, Mathieu Coureuil, Xavier Nassif, Mohamed El Behi.

References

- Stephens DS. Biology and pathogenesis of the evolutionarily successful, obligate human bacterium *Neisseria meningitidis*. Vaccine. 2009; 27: B71–B77. https://doi.org/10.1016/j.vaccine.2009.04.070 PMID: 19477055
- 2. Audry M, Robbe-Masselot C, Barnier J-P, Gachet B, Saubaméa B, Schmitt A, et al. Airway Mucus Restricts *Neisseria meningitidis* Away from Nasopharyngeal Epithelial Cells and Protects the Mucosa from Inflammation. mSphere. 2019; 4. https://doi.org/10.1128/mSphere.00494-19 PMID: 31801841
- Christensen H, May M, Bowen L, Hickman M, Trotter CL. Meningococcal carriage by age: a systematic review and meta-analysis. Lancet Infect Dis. 2010; 10: 853–861. https://doi.org/10.1016/S1473-3099 (10)70251-6 PMID: 21075057
- Bernard SC, Simpson N, Join-Lambert O, Federici C, Laran-Chich M-P, Maïssa N, et al. Pathogenic Neisseria meningitidis utilizes CD147 for vascular colonization. Nat Med. 2014; 20: 725–31. https://doi. org/10.1038/nm.3563 PMID: 24880614
- Capel E, Barnier J-P, Zomer AL, Bole-Feysot C, Nussbaumer T, Jamet A, et al. Peripheral blood vessels are a niche for blood-borne meningococci. Virulence. 2017; 8: 1808–1819. https://doi.org/10.1080/ 21505594.2017.1391446 PMID: 29099305
- Join-Lambert O, Lecuyer H, Miller F, Lelievre L, Jamet A, Furio L, et al. Meningococcal Interaction to Microvasculature Triggers the Tissular Lesions of Purpura Fulminans. J Infect Dis. 2013; 208: 1590– 1597. https://doi.org/10.1093/infdis/jit301 PMID: 23840047
- Melican K, Dumenil G. Vascular colonization by *Neisseria meningitidis*. Curr Opin Microbiol. 2012; 15: 50–56. https://doi.org/10.1016/j.mib.2011.10.008 PMID: 22185907
- Denis K, Le Bris M, Le Guennec L, Barnier J-P, Faure C, Gouge A, et al. Targeting Type IV pili as an antivirulence strategy against invasive meningococcal disease. Nat Microbiol. 2019. https://doi.org/10. 1038/s41564-019-0395-8 PMID: 30911127
- Lécuyer H, Virion Z, Barnier J-P, Matczak S, Bourdoulous S, Bianchini E, et al. An ADAM-10 dependent EPCR shedding links meningococcal interaction with endothelial cells to purpura fulminans. Luo Z-Q, editor. PLOS Pathog. 2018; 14: e1006981. <u>https://doi.org/10.1371/journal.ppat.1006981</u> PMID: 29630665
- Brandtzaeg P, Deuren M. Classification and Pathogenesis of Meningococcal Infections. In: Christodoulides M, editor. *Neisseria meningitidis*. Totowa, NJ: Humana Press; 2012. pp. 21–35. Available: <u>http://</u> www.springerlink.com/index/10.1007/978-1-61779-346-2_2
- Sotto MN, Langer B, Hoshino-Shimizu S, de Brito T. Pathogenesis of Cutaneous Lesions in Acute Meningococcemia in Humans: Light, Immunofluorescent, and Electron Microscopic Studies of Skin Biopsy Specimens. J Infect Dis. 1976; 133: 506–514. <u>https://doi.org/10.1093/infdis/133.5.506</u> PMID: 816975
- Staquet P, Lemee L, Verdier E, Bonmarchand G, Laudenbach V, Michel C, et al. Detection of *Neisseria meningitidis* DNA from skin lesion biopsy using real-time PCR: usefulness in the aetiological diagnosis of purpura fulminans. Intensive Care Med. 2007; 33: 1168–1172. <u>https://doi.org/10.1007/s00134-007-0667-2 PMID</u>: 17503017
- Thompson MJ, Ninis N, Perera R, Mayon-White R, Phillips C, Bailey L, et al. Clinical recognition of meningococcal disease in children and adolescents. The Lancet. 2006; 367: 397–403. <u>https://doi.org/ 10.1016/S0140-6736(06)67932-4</u> PMID: 16458763
- Beau C, Vlassova N, Sarlangue J, Brissaud O, Léauté-Labrèze C, Boralevi F. Diagnostic Value of Polymerase Chain Reaction Analysis of Skin Biopsies in Purpura Fulminans. Pediatr Dermatol. 2013; 30: e276–e277. https://doi.org/10.1111/pde.12181 PMID: 23834254
- Darton T, Guiver M, Naylor S, Jack DL, Kaczmarski EB, Borrow R, et al. Severity of Meningococcal Disease Associated with Genomic Bacterial Load. Clin Infect Dis. 2009; 48: 587–594. <u>https://doi.org/10.1086/596707 PMID: 19191644</u>
- Hackett SJ, Guiver M, Marsh J, Sills JA, Thomson APJ, Kaczmarski EB, et al. Meningococcal bacterial DNA load at presentation correlates with disease severity. Arch Dis Child. 2002; 86: 44–46. <u>https://doi.org/10.1136/adc.86.1.44</u> PMID: 11806883
- Dupin N, Lecuyer H, Carlotti A, Poyart C, Coureuil M, Chanal J, et al. Chronic Meningococcemia Cutaneous Lesions Involve Meningococcal Perivascular Invasion Through the Remodeling of Endothelial Barriers. Clin Infect Dis. 2012; 54: 1162–1165. https://doi.org/10.1093/cid/cis120 PMID: 22412064
- Lefèvre B, Piau C, Javaugue F-C, Talarmin J-P, Lefebvre M, Varache N, et al. Chronic meningococcemia: a report of 26 cases and literature review. Infection. 2019; 47: 285–288. <u>https://doi.org/10.1007/s15010-018-1238-x PMID: 30341638</u>

- Nassif X, Lowy J, Stenberg P, O'Gaora P, Ganji A, So M. Antigenic variation of pilin regulates adhesion of *Neisseria meningitidis* to human epithelial cells. Mol Microbiol. 1993; 8: 719–725. https://doi.org/10. 1111/j.1365-2958.1993.tb01615.x PMID: 8332064
- Nassif X, Marceau M, Pujol C, Pron B, Beretti J-L, Taha M-K. Type-4 pili and meningococcal adhesiveness. Gene. 1997; 192: 149–153. https://doi.org/10.1016/s0378-1119(96)00802-5 PMID: 9224885
- Virji M, Saunders JR, Sims G, Makepeace K, Maskell D, Ferguson DJP. Pilus-facilitated adherence of Neisseria meningitidis to human epithelial and endothelial cells: modulation of adherence phenotype occurs concurrently with changes in primary amino acid sequence and the glycosylation status of pilin. Mol Microbiol. 1993; 10: 1013–1028. https://doi.org/10.1111/j.1365-2958.1993.tb00972.x PMID: 7934852
- Berry J-L, Cehovin A, McDowell MA, Lea SM, Pelicic V. Functional Analysis of the Interdependence between DNA Uptake Sequence and Its Cognate ComP Receptor during Natural Transformation in *Neisseria* Species. Hughes D, editor. PLoS Genet. 2013; 9: e1004014. <u>https://doi.org/10.1371/journal.pgen.1004014</u> PMID: 24385921
- Cehovin A, Simpson PJ, McDowell MA, Brown DR, Noschese R, Pallett M, et al. Specific DNA recognition mediated by a type IV pilin. Proc Natl Acad Sci. 2013; 110: 3065–3070. <u>https://doi.org/10.1073/pnas.1218832110</u> PMID: 23386723
- Helaine S, Carbonnelle E, Prouvensier L, Beretti J-L, Nassif X, Pelicic V. PilX, a pilus-associated protein essential for bacterial aggregation, is a key to pilus-facilitated attachment of *Neisseria meningitidis* to human cells. Mol Microbiol. 2004; 55: 65–77. https://doi.org/10.1111/j.1365-2958.2004.04372.x PMID: 15612917
- Merz AJ, So M, Sheetz MP. Pilus retraction powers bacterial twitching motility. Nature. 2000; 407: 98– 102. https://doi.org/10.1038/35024105 PMID: 10993081
- Wolfgang M, Lauer P, Park H-S, Brossay L, Hébert J, Koomey M. PilT mutations lead to simultaneous defects in competence for natural transformation and twitching motility in piliated *Neisseria gonorrhoeae*. Mol Microbiol. 1998; 29: 321–330. https://doi.org/10.1046/j.1365-2958.1998.00935.x PMID: 9701824
- Melican K, Michea Veloso P, Martin T, Bruneval P, Duménil G. Adhesion of *Neisseria meningitidis* to Dermal Vessels Leads to Local Vascular Damage and Purpura in a Humanized Mouse Model. Tang C, editor. PLoS Pathog. 2013; 9: e1003139. https://doi.org/10.1371/journal.ppat.1003139 PMID: 23359320
- Pron B, Taha M-K, Rambaud C, Fournet J-C, Pattey N, Monnet J-P, et al. Interaction of *Neisseria meningitidis* with the Components of the Blood-Brain Barrier Correlates with an Increased Expression of PilC. J Infect Dis. 1997; 176: 1285–1292. https://doi.org/10.1086/514124 PMID: 9359730
- 29. Craig L, Pique ME, Tainer JA. Type IV pilus structure and bacterial pathogenicity. Nat Rev Microbiol. 2004; 2: 363–378. https://doi.org/10.1038/nrmicro885 PMID: 15100690
- Kolappan S, Coureuil M, Yu X, Nassif X, Egelman EH, Craig L. Structure of the Neisseria meningitidis Type IV pilus. Nat Commun. 2016; 7: 13015. https://doi.org/10.1038/ncomms13015 PMID: 27698424
- Helaine S, Dyer DH, Nassif X, Pelicic V, Forest KT. 3D structure/function analysis of PilX reveals how minor pilins can modulate the virulence properties of type IV pili. Proc Natl Acad Sci. 2007; 104: 15888– 15893. https://doi.org/10.1073/pnas.0707581104 PMID: 17893339
- Pelicic V. Type IV pili: e pluribus unum? Mol Microbiol. 2008; 68: 827–837. <u>https://doi.org/10.1111/j.</u> 1365-2958.2008.06197.x PMID: 18399938
- Morand PC, Tattevin P, Eugene E, Beretti J-L, Nassif X. The adhesive property of the type IV pilusassociated component PilC1 of pathogenic *Neisseria* is supported by the conformational structure of the N-terminal part of the molecule. Mol Microbiol. 2001; 40: 846–856. https://doi.org/10.1046/j.1365-2958.2001.02452.x PMID: 11401692
- Morand PC, Bille E, Morelle S, Eugène E, Beretti J-L, Wolfgang M, et al. Type IV pilus retraction in pathogenic *Neisseria* is regulated by the PilC proteins. EMBO J. 2004; 23: 2009–2017. https://doi.org/10. 1038/sj.emboj.7600200 PMID: 15103324
- Nassif X, Beretti JL, Lowy J, Stenberg P, O'Gaora P, Pfeifer J, et al. Roles of pilin and PilC in adhesion of *Neisseria meningitidis* to human epithelial and endothelial cells. Proc Natl Acad Sci U S A. 1994; 91: 3769–3773. https://doi.org/10.1073/pnas.91.9.3769 PMID: 7909606
- Rudel T, Scheuerpflug I, Meyer TF. Neisseria PilC protein identified as type-4 pilus tip-located adhesin. Nature. 1995; 373: 357–359. https://doi.org/10.1038/373357a0 PMID: 7830772
- Yasukawa K, Martin P, Tinsley CR, Nassif X. Pilus-mediated adhesion of *Neisseria meningitidis* is negatively controlled by the pilus-retraction machinery: PilT downregulates the expression of pilC1. Mol Microbiol. 2006; 59: 579–589. https://doi.org/10.1111/j.1365-2958.2005.04954.x PMID: 16390451

- Pujol C, Eugène E, Marceau M, Nassif X. The meningococcal PilT protein is required for induction of intimate attachment to epithelial cells following pilus-mediated adhesion. Proc Natl Acad Sci. 1999; 96: 4017–4022. https://doi.org/10.1073/pnas.96.7.4017 PMID: 10097155
- Biais N, Ladoux B, Higashi D, So M, Sheetz M. Cooperative retraction of bundled Type IV Pili enables nanonewton force generation. Schliwa M, editor. PLoS Biol. 2008; 6: e87. https://doi.org/10.1371/ journal.pbio.0060087 PMID: 18416602
- Bonazzi D, Lo Schiavo V, Machata S, Djafer-Cherif I, Nivoit P, Manriquez V, et al. Intermittent Pili-Mediated Forces Fluidize *Neisseria meningitidis* Aggregates Promoting Vascular Colonization. Cell. 2018; 174: 143–155.e16. https://doi.org/10.1016/j.cell.2018.04.010 PMID: 29779947
- Lo H, Tang CM, Exley RM. Mechanisms of avoidance of host immunity by Neisseria meningitidis and its effect on vaccine development. Lancet Infect Dis. 2009; 9: 418–427. <u>https://doi.org/10.1016/S1473-3099(09)70132-X PMID: 19555901</u>
- Nolte MA, Arens R, Kraus M, Oers MHJ van, Kraal G, Lier RAW van, et al. B Cells Are Crucial for Both Development and Maintenance of the Splenic Marginal Zone. J Immunol. 2004; 172: 3620–3627. https://doi.org/10.4049/jimmunol.172.6.3620 PMID: 15004164
- Willems L, Li S, Rutgeerts O, Lenaerts C, Waer M, Billiau AD. IL-7 Is Required for the Development of the Intrinsic Function of Marginal Zone B Cells and the Marginal Zone Microenvironment. J Immunol. 2011; 187: 3587–3594. https://doi.org/10.4049/jimmunol.1004012 PMID: 21873520
- You Y, Zhao H, Wang Y, Carter RH. Cutting Edge: Primary and Secondary Effects of CD19 Deficiency on Cells of the Marginal Zone. J Immunol. 2009; 182: 7343–7347. <u>https://doi.org/10.4049/jimmunol.</u> 0804295 PMID: 19494255
- Hockenberry AM, Hutchens DM, Agellon A, So M. Attenuation of the Type IV Pilus Retraction Motor Influences Neisseria gonorrhoeae Social and Infection Behavior. mBio. 2016; 7. <u>https://doi.org/10.</u> 1128/mBio.01994-16 PMID: 27923924
- 46. Mairey E, Duménil G. Cerebral microcirculation shear stress levels determine Neisseria meningitidis attachment sites along the blood-brain barrier. J Exp Med. 2006; 203: 1939–1950. https://doi.org/10. 1084/jem.20060482 PMID: 16864659
- Aukema KG, Kron EM, Herdendorf TJ, Forest KT. Functional Dissection of a Conserved Motif within the Pilus Retraction Protein PilT. J Bacteriol. 2005; 187: 611–618. <u>https://doi.org/10.1128/JB.187.2.611-618.2005</u> PMID: 15629932
- Craig L, Forest KT, Maier B. Type IV pili: dynamics, biophysics and functional consequences. Nat Rev Microbiol. 2019; 17: 429–440. https://doi.org/10.1038/s41579-019-0195-4 PMID: 30988511
- Pelicic V. Monoderm bacteria: the new frontier for type IV pilus biology. Mol Microbiol. 2019; 112: 1674– 1683. https://doi.org/10.1111/mmi.14397 PMID: 31556183
- Bieber D, Ramer SW, Wu C-Y, Murray WJ, Tobe T, Fernandez R, et al. Type IV Pili, Transient Bacterial Aggregates, and Virulence of Enteropathogenic *Escherichia coli*. Science. 1998; 280: 2114–2118. https://doi.org/10.1126/science.280.5372.2114 PMID: 9641917
- Comolli JC, Hauser AR, Waite L, Whitchurch CB, Mattick JS, Engel JN. *Pseudomonas aeruginosa* Gene Products PilT and PilU Are Required for Cytotoxicity In Vitro and Virulence in a Mouse Model of Acute Pneumonia. Infect Immun. 1999; 67: 3625–3630. <u>https://doi.org/10.1128/IAI.67.7.3625-3630</u>. 1999 PMID: 10377148
- Eriksson J, Eriksson OS, Jonsson A-B. Loss of Meningococcal PilU Delays Microcolony Formation and Attenuates Virulence In Vivo. Infect Immun. 2012; 80: 2538–2547. https://doi.org/10.1128/IAI.06354-11 PMID: 22508857
- Han X, Kennan RM, Davies JK, Reddacliff LA, Dhungyel OP, Whittington RJ, et al. Twitching Motility Is Essential for Virulence in *Dichelobacter nodosus*. J Bacteriol. 2008; 190: 3323–3335. <u>https://doi.org/10.1128/JB.01807-07</u> PMID: 18310333
- Sexton JA, Pinkner JS, Roth R, Heuser JE, Hultgren SJ, Vogel JP. The Legionella pneumophila PiIT Homologue DotB Exhibits ATPase Activity That Is Critical for Intracellular Growth. J Bacteriol. 2004; 186: 1658–1666. https://doi.org/10.1128/jb.186.6.1658-1666.2004 PMID: 14996796
- Johansson L. CD46 in Meningococcal Disease. Science. 2003; 301: 373–375. <u>https://doi.org/10.1126/</u> science.1086476 PMID: 12869763
- 56. Kirchner M, Heuer D, Meyer TF. CD46-Independent Binding of Neisserial Type IV Pili and the Major Pilus Adhesin, PilC, to Human Epithelial Cells. Infect Immun. 2005; 73: 3072–3082. <u>https://doi.org/10.1128/IAI.73.5.3072-3082.2005 PMID: 15845515</u>
- 57. Coureuil M, Lécuyer H, Scott MGH, Boularan C, Enslen H, Soyer M, et al. Meningococcus Hijacks a β2-Adrenoceptor/β-Arrestin Pathway to Cross Brain Microvasculature Endothelium. Cell. 2010; 143: 1149–1160. https://doi.org/10.1016/j.cell.2010.11.035 PMID: 21183077

- 58. Virion Z, Doly S, Saha K, Lambert M, Guillonneau F, Bied C, et al. Sialic acid mediated mechanical activation of β 2 adrenergic receptors by bacterial pili. Nat Commun. 2019; 10: 4752. https://doi.org/10. 1038/s41467-019-12685-6 PMID: 31628314
- 59. Welker A, Cronenberg T, Zöllner R, Meel C, Siewering K, Bender N, et al. Molecular Motors Govern Liquidlike Ordering and Fusion Dynamics of Bacterial Colonies. Phys Rev Lett. 2018; 121: 118102. https://doi.org/10.1103/PhysRevLett.121.118102 PMID: 30265121
- Zöllner R, Cronenberg T, Kouzel N, Welker A, Koomey M, Maier B. Type IV Pilin Post-Translational Modifications Modulate Material Properties of Bacterial Colonies. Biophys J. 2019; 116: 938–947. https://doi.org/10.1016/j.bpj.2019.01.020 PMID: 30739725
- Carbonnelle E, Hill DJ, Morand P, Griffiths NJ, Bourdoulous S, Murillo I, et al. Meningococcal interactions with the host. Vaccine. 2009; 27: B78–B89. https://doi.org/10.1016/j.vaccine.2009.04.069 PMID: 19481311
- 62. Bradley CJ, Griffiths NJ, Rowe HA, Heyderman RS, Virji M. Critical determinants of the interactions of capsule-expressing *Neisseria meningitidis* with host cells: the role of receptor density in increased cellular targeting via the outer membrane Opa proteins. Cell Microbiol. 2005; 7: 1490–1503. https://doi.org/ 10.1111/j.1462-5822.2005.00572.x PMID: 16153247
- Caugant DA. Genetics and evolution of *Neisseria meningitidis*: Importance for the epidemiology of meningococcal disease. Infect Genet Evol. 2008; 8: 558–565. https://doi.org/10.1016/j.meegid.2008. 04.002 PMID: 18479979
- Merz AJ, So M. Attachment of piliated, Opa-and Opc-gonococci and meningococci to epithelial cells elicits cortical actin rearrangements and clustering of tyrosine-phosphorylated proteins. Infect Immun. 1997; 65: 4341–4349. https://doi.org/10.1128/IAI.65.10.4341-4349.1997 PMID: 9317047
- Cunha CSE, Griffiths NJ, Virji M. Neisseria meningitidis Opc invasin binds to the sulphated tyrosines of activated vitronectin to attach to and invade human brain endothelial cells. PLOS Pathog. 2010; 6: e1000911. https://doi.org/10.1371/journal.ppat.1000911 PMID: 20502634
- Unkmeir A, Latsch K, Dietrich G, Wintermeyer E, Schinke B, Schwender S, et al. Fibronectin mediates Opc-dependent internalization of *Neisseria meningitidis* in human brain microvascular endothelial cells. Mol Microbiol. 2002; 46: 933–946. https://doi.org/10.1046/j.1365-2958.2002.03222.x PMID: 12421301
- Virji M, Makepeace K, Ferguson DJP, Achtman M, Sarkari J, Moxon ER. Expression of the Opc protein correlates with invasion of epithelial and endothelial cells by *Neisseria meningitidis*. Mol Microbiol. 1992; 6: 2785–2795. https://doi.org/10.1111/j.1365-2958.1992.tb01458.x PMID: 1435257
- Seiler A, Reinhardt R, Sarkari J, Caugant DA, Achtman M. Allelic polymorphism and site-specific recombination in the opc locus of *Neisseria meningitidis*. Mol Microbiol. 1996; 19: 841–856. <u>https://doi.org/10.1046/j.1365-2958.1996.437970.x PMID: 8820653</u>
- **69.** Whalen CM, Hockin JC, Ryan A, Ashton F. The Changing Epidemiology of Invasive Meningococcal Disease in Canada, 1985 Through 1992: Emergence of a Virulent Clone of *Neisseria meningitidis*. JAMA. 1995; 273: 390–394. https://doi.org/10.1001/jama.1995.03520290042027 PMID: 7823384
- 70. van der Woude MW, Baumler AJ. Phase and Antigenic Variation in Bacteria. Clin Microbiol Rev. 2004; 17: 581–611. https://doi.org/10.1128/CMR.17.3.581-611.2004 PMID: 15258095
- Green LR, Dave N, Adewoye AB, Lucidarme J, Clark SA, Oldfield NJ, et al. Potentiation of Phase Variation in Multiple Outer-Membrane Proteins During Spread of the Hyperinvasive Neisseria meningitidis Serogroup W ST-11 Lineage. J Infect Dis. 2019; 220: 1109–1117. https://doi.org/10.1093/infdis/jiz275 PMID: 31119276
- 72. Rytkönen A, Albiger B, Hansson-Palo P, Källström H, Olcén P, Fredlund H, et al. Neisseria meningitidis Undergoes PilC Phase Variation and PilE Sequence Variation during Invasive Disease. J Infect Dis. 2004; 189: 402–409. https://doi.org/10.1086/381271 PMID: 14745697
- 73. Chamot-Rooke J, Mikaty G, Malosse C, Soyer M, Dumont A, Gault J, et al. Posttranslational Modification of Pili upon Cell Contact Triggers N. meningitidis Dissemination. Science. 2011; 331: 778–782. https://doi.org/10.1126/science.1200729 PMID: 21311024
- 74. Sigurlásdóttir S, Engman J, Eriksson OS, Saroj SD, Zguna N, Lloris-Garcerá P, et al. Host cell-derived lactate functions as an effector molecule in *Neisseria meningitidis* microcolony dispersal. Tang C, editor. PLOS Pathog. 2017; 13: e1006251. https://doi.org/10.1371/journal.ppat.1006251 PMID: 28384279
- 75. Dewenter L, Volkmann TE, Maier B. Oxygen governs gonococcal microcolony stability by enhancing the interaction force between type IV pili. Integr Biol. 2015; 7: 1161–1170. https://doi.org/10.1039/ c5ib00018a PMID: 25892255
- 76. Ovstebo R, Brandtzaeg P, Brusletto B, Haug KBF, Lande K, Høiby EA, et al. Use of Robotized DNA Isolation and Real-Time PCR To Quantify and Identify Close Correlation between Levels of *Neisseria meningitidis* DNA and Lipopolysaccharides in Plasma and Cerebrospinal Fluid from Patients with Systemic Meningococcal Disease. J Clin Microbiol. 2004; 42: 2980–2987. <u>https://doi.org/10.1128/JCM.42</u>. 7.2980-2987.2004 PMID: 15243048

- Hodeib S, Herberg JA, Levin M, Sancho-Shimizu V. Human genetics of meningococcal infections. Hum Genet. 2020; 139: 961–980. https://doi.org/10.1007/s00439-020-02128-4 PMID: 32067109
- Lewis LA, Ram S. Meningococcal disease and the complement system. Virulence. 2014; 5: 98–126. https://doi.org/10.4161/viru.26515 PMID: 24104403
- 79. Unkmeir A, Kämmerer U, Stade A, Hübner C, Haller S, Kolb-Mäurer A, et al. Lipooligosaccharide and Polysaccharide Capsule: Virulence Factors of *Neisseria meningitidis* That Determine Meningococcal Interaction with Human Dendritic Cells. Infect Immun. 2002; 70: 2454–2462. <u>https://doi.org/10.1128/iai.</u> 70.5.2454-2462.2002 PMID: 11953382
- 80. van Deuren M, van der Ven Jongekrijg J, Demacker PNM, Bartelink AKM, van Dalen R, Sauerwein RW, et al. Differential Expression of Proinflammatory Cytokines and Their Inhibitors during the Course of Meningococcal Infections. J Infect Dis. 1994; 169: 157–161. https://doi.org/10.1093/infdis/169.1.157 PMID: 8277177
- van Deuren M, van der Ven-Jongekrijg J, Bartelink AKM, van Dalen R, Sauerwein RW, van der Meer JWM. Correlation between Proinflammatory Cytokines and Antiinflammatory Mediators and the Severity of Disease in Meningococcal Infections. J Infect Dis. 1995; 172: 433–439. https://doi.org/10.1093/ infdis/172.2.433 PMID: 7622886
- Girardin E, Grau GE, Dayer J-M, Roux-Lombard P, Lambert P-H. Tumor Necrosis Factor and Interleuktn-1 in the Serum of Children with Severe Infectious Purpura. N Engl J Med. 1988; 319: 397–400. https://doi.org/10.1056/NEJM198808183190703 PMID: 3135497
- Van Deuren M, Brandtzaeg P, van der Meer JW. Update on meningococcal disease with emphasis on pathogenesis and clinical management. Clin Microbiol Rev. 2000; 13: 144–166. <u>https://doi.org/10.1128/ cmr.13.1.144-166.2000</u> PMID: 10627495
- Bergounioux J, Coureuil M, Belli E, Ly M, Cambillau M, Goudin N, et al. Experimental Evidence of Bacterial Colonization of Human Coronary Microvasculature and Myocardial Tissue during Meningococcemia. Infect Immun. 2016; 84: 3017–3023. https://doi.org/10.1128/IAI.00420-16 PMID: 27481255
- Guarner J, Greer PW, Whitney A, Shieh W-J, Fischer M, White EH, et al. Pathogenesis and Diagnosis of Human Meningococcal Disease Using Immunohistochemical and PCR Assays. Am J Clin Pathol. 2004; 122: 754–764. https://doi.org/10.1309/A7M2-FN2T-YE6A-8UFX PMID: 15491972
- Makwana N, Baines PB. Myocardial dysfunction in meningococcal septic shock. Curr Opin Crit Care. 2005; 11: 418–423. https://doi.org/10.1097/01.ccx.0000176699.51456.13 PMID: 16175027
- Cheng S-C, Scicluna BP, Arts RJW, Gresnigt MS, Lachmandas E, Giamarellos-Bourboulis EJ, et al. Broad defects in the energy metabolism of leukocytes underlie immunoparalysis in sepsis. Nat Immunol. 2016; 17: 406–413. https://doi.org/10.1038/ni.3398 PMID: 26950237
- Delano MJ, Ward PA. The immune system's role in sepsis progression, resolution, and long-term outcome. Immunol Rev. 2016; 274: 330–353. https://doi.org/10.1111/imr.12499 PMID: 27782333
- Reyes M, Filbin MR, Bhattacharyya RP, Billman K, Eisenhaure T, Hung DT, et al. An immune-cell signature of bacterial sepsis. Nat Med. 2020; 26: 333–340. <u>https://doi.org/10.1038/s41591-020-0752-4</u> PMID: 32066974
- Ramsey ME, Hackett KT, Kotha C, Dillard JP. New Complementation Constructs for Inducible and Constitutive Gene Expression in *Neisseria gonorrhoeae* and *Neisseria meningitidis*. Appl Environ Microbiol. 2012; 78: 3068–3078. https://doi.org/10.1128/AEM.07871-11 PMID: 22327577